Author’s response to reviews

Title: Analysis of factors influencing the frequency of primary care visits among diabetic patients in two provinces in China

Authors:

JinWen Wang (wang-jin-wen@163.com)
Huijuan Zuo (huijuezuo@sina.com)
XiaoRong Chen (xiaorongch@126.com)
Lei Hou (houlei_houlei@126.com)
JiXiang Ma (jixiangma618@163.com)

Version: 3 Date: 30 Aug 2019

Author’s response to reviews:

We are very pleased to know that our manuscript has been received, thank you very much for the work and help.

Response to the editor

1. In your “Ethics approval and consent to participate” section of your Declarations, please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state it in this section. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

The authors declare that written consent was obtained from all participants.

2. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The funding bodies had no role in the design of the study, collection, analysis, and interpretation of data or in writing the manuscript.
3. Thank you for uploading your cover letter. As this file is no longer required at this stage of the editorial process, please remove it from your submission.

Cover letter has been removed.

4. We note that you have not cited your supplementary tables within the text. Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

We have cited the additional tables within the text.

5. Please upload your revised manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.

The revised manuscript has been uploaded as a single, final, clean version.

Once again, the hard work of editors is very much appreciated.

Yours faithfully,

Jinwen Wang, PHD

Beijing Anzhen Hospital, Capital Medical University, Beijing Institute of Heart Lung and Blood Vessel Disease, Beijing 100029, China

Email: wang-jin-wen@163.com